Author's response to reviews

Title: Imatinib Induced Severe Skin Reactions and Neutropenia in a Patient with Gastrointestinal Stromal Tumor

Authors:

Jun-Eul Hwang (hjunyl@naver.com)
Ju-Young Yoon (flanaria107@hanmail.net)
Woo-Kyun Bae (drwookyun@hanmail.net)
Hyun-Jeong Shim (hjnhj@hanmail.net)
Sang-Hee Cho (sh115@chollian.net)
Ik-Joo Chung (ijchung@chonnam.ac.kr)

Version: 4 Date: 14 August 2010

Author's response to reviews: see over
Dear Editor

We have enclosed a revised manuscript entitled “Imatinib Induced Severe Skin Reactions and Neutropenia in a Patient with Gastrointestinal Stromal Tumor” which we would like to submit to "BMC cancer" for consideration as a case report. We would be grateful if the manuscript could be reviewed and considered for publication in “BMC cancer”.

We are very inspired and thank you for your review.

Sincerely yours,

Ik-Joo Chung, M.D., Ph.D.

Corresponding author and request for reprints:

Ik-Joo Chung, M.D, Ph.D.
Department of Internal Medicine,
Chonnam National University Hwasun Hospital,
160 Ilsim-ri, Hwasun-eup, Hwasun-gun 519-809, Korea
Telephone: +82-61-379-7632; Fax: +82-61-379-7628
E-mail: ijchung@chonnam.ac.kr
Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format:
Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

→

Affiliation: ¹Department of Hematology-Oncology, Chonnam National University Medical School, Gwangju 501-757, South Korea.

²The Brain Korea 21 Project, Center for Biomedical Human Resources, Jeonnam Regional Cancer Center, Chonnam National University Hwasun Hospital, Hwasun, Jeollanam-do, South Korea.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Acknowledgements: please remove this repeated section.

→ I removed this repeated section.